Author's response to reviews

Title: Downregulation of Toll-like receptor 4 induces suppressive effects on hepatitis B virus-related hepatocellular carcinoma via ERK1/2 signaling

Authors:

Yiting Wang (yitingwang_1111@163.com)
Jing Cai (cjdl123@163.com)
Xiaoli Zeng (rememberme0915@126.com)
Yajie Chen (yichen0506@163.com)
Wei Yan (yanweishandian@126.com)
Yuming Ouyang (oyym89@163.com)
Dan Xiao (xd870824@163.com)
Zhimin Zeng (Zzm@163.com)
Long Huang (huanglong914@gmail.com)
Anwen Liu (awliu666@163.com)

Version: 8
Date: 20 October 2015

Author's response to reviews: see over
Cover letter

It was our mistake that we put a duplicate record of Dan Xiao in the authors section. In the revised manuscript, we removed this duplicate record from title page and everything is good now. Thank you for your instruction.